Author’s response to reviews

Title: Upper airway stabilization by osteopathic manipulation of the sphenopalatine ganglion versus sham manipulation in OSAS patients: a proof-of-concept, randomized, crossover, double-blind, controlled study

Authors:

Olivier Jacq (jacqolivier.osteo@gmail.com)
Isabelle Arnulf (isabelle.arnulf@aphp.fr)
Thomas Similowski (thomas.similowski@aphp.fr)
valerie attali (valerie.attali@aphp.fr)

Version: 5 Date: 05 Dec 2017

Author’s response to reviews:

Dear Editor,

Please find our comments here below:

1. Thank you for revising your ethics statement in your response letter, but this change has not been made in your actual manuscript. Please make that change.

Answer: Thank you for this remark. We apologize, we had changed in the manuscript but not at the section "ethic". The correction was made.

2. Please move the statement, "This research was legally sponsored by the non-profit association ADOREPS ("Association pour le Développement et l’Organisation de la Recherche en Pneumologie et sur le Sommeil") from the Acknowledgement section to the Funding section.

Answer: this was done. However We are not able to complete the funding informations that is required in the submission system (grant numer etc...), as we had no fees and consequently no grant number. Let me know if I have to complete something.

Many thanks
Best Regards

Dr Valérie Attali